Author’s response to reviews

Title: Perceptions of HIV/STI prevention among young adults in Sweden who travel abroad. A qualitative study with focus group and individual interviews

Authors:

Anna Qvarnström (anna.qvarnstrom@lnu.se)
Marie G Oscarsson (marie.oscarsson@lnu.se)

Version: 4 Date: 19 August 2014

Author’s response to reviews: see over
Dear Editorial Board of BMC Public Health,

On behalf of both authors, I hereby submit the revised manuscript ID: 3036878601220784, title “Perceptions of HIV/STI prevention among young adults in Sweden who travel abroad. A qualitative study with focus group and individual interviews”.

We are very pleased to hear that the manuscript is close to acceptance, and that the reviewers are satisfied with the revised version of the manuscript. In our coming studies on this topic, we will bear in mind the recommendations from Referee 1, that a more interpretative analysis could add more depth to the analysis and give the reader interesting perspectives on prevention issues for those who travel abroad. We believe that this explorative study makes a good starting point for future studies within this topic.

According to the editorial request, we have ensured that our study adheres to the RATS guidelines for reporting qualitative studies. We have included a sentence stating this at the end of the Methods section. We have also made sure that the revised manuscript conforms to the journal style and is correctly formatted.

We hope you find the revised version satisfying.

Best regards,

Anna Qvarnström